Colon injury after blunt abdominal trauma is generally occurred as a result of in- or out-vehicle accidents, falling from high and assault. The number of cases with colon injury due to minor trauma, as in our case (the patient fell in a bathroom and hit his/her abdomen), is very limited. However, the most tragic situation is paying no attention to these kinds of traumas by the patient and his/her relatives or ignoring until sepsis developed and thereby not presenting to the hospital. In these cases, in whom fecal contamination is very high, morbidity and mortality rates are increased. One of the aims of the present study is to draw attention to these kinds of traumas which are ignored by the patients.